Author's response to reviews

Title: Inhaled Nitric Oxide for the Adjunctive Therapy of Severe Malaria: Protocol for a Randomized Controlled Trial

Authors:

Michael Hawkes (michael.hawkes@utoronto.ca)
Robert Opika Opoka (opokabob@yahoo.com)
Sophie Namasopo (snamasopo@yahoo.com)
Christopher Miller (miller42@interchange.ubc.ca)
Kevin E Thorpe (kevin.thorpe@utoronto.ca)
James V Lavery (jim.lavery@mrcglobal.org)
Andrea L Conroy (andrea.conroy@gmail.com)
Wayne C Liles (conrad.liles@uhn.on.ca)
Chandy C John (ccj@umn.edu)
Kevin C Kain (kevin.kain@uhn.on.ca)

Version: 2 Date: 16 February 2011

Author's response to reviews:

Dear Ragav Sawhney and the Trials Editorial Team,

We are attaching the requested documentation of ethical approval:

1. Makerere University, Kampala, Uganda
2. University of Toronto, Canada
3. National Drug Authority, Uganda

as well as documentation of trial funding:

1. Memorandum of Understanding, McLaughlin Rotman Centre for Global Health and University Health Network.

Thank you and we look forward to your review of our manuscript,

Michael Hawkes and Kevin Kain for the co-authors